Author's response to reviews

Title: Echocardiographic Partition Values and Prevalence of Left Ventricular Hypertrophy in Hypertensive Nigerians.

Authors:

Adewole Adebiyi (wadebiyi@yahoo.com)
Okechukwu S Ogah (osogah56156@yahoo.com)
Akinyemi Aje (ajeyemi@yahoo.co.uk)
Dike B Ojji (dikeojji@yahoo.co.uk)
Adedeji K Adebayo (akbayo53972@yahoo.com)
Olulola O Oladapo (dowo@skannet.com.ng)
Ayodele O Falase (aofalase@yahoo.com)

Version: 3 Date: 19 August 2006

Author's response to reviews:

1. Abstract: The first subsection has been renamed Background and the last Conclusions
2. The background section of the abstract has been updated. This has been reflected in the submission system.
3. Introduction: This has been changed to Background
4. Information on consent and ethical approval has been included
5. Conclusion has been changed to conclusions